Early versus deferred endovenous ablation of superficial venous reflux in patients with venous ulceration: the EVRA RCT

Manjit S Gohel,^{1,2} Francine Heatley,² Xinxue Liu,³ Andrew Bradbury,⁴ Richard Bulbulia,^{5,6,7} Nicky Cullum,⁸ David M Epstein,⁹ Isaac Nyamekye,¹⁰ Keith R Poskitt,⁵ Sophie Renton,¹¹ Jane Warwick^{3,12} and Alun H Davies^{2*} on behalf of the EVRA trial investigators

 ¹Cambridge University Hospitals NHS Foundation Trust, Cambridge, UK
²Department of Surgery and Cancer, Imperial College London, London, UK
³Imperial Clinical Trials Unit, Imperial College London, London, UK
⁴College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK
⁵Gloucestershire Hospitals NHS Foundation Trust, Gloucester, UK
⁶Medical Research Council Population Health Research Unit, Nuffield Department of Population Health, University of Oxford, Oxford, UK
⁷Clinical Trial Service Unit and Epidemiological Studies Unit, Nuffield Department of Population Health, University of Oxford, Oxford, UK
⁸School of Health Sciences, University of Manchester, Manchester, UK
⁹Department of Applied Economics, University of Granada, Granada, Spain
¹⁰Worcestershire Acute Hospitals NHS Trust, London, UK
¹²Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, Coventry, UK

*Corresponding author a.h.davies@imperial.ac.uk

Declared competing interests of authors: Manjit S Gohel has received personal fees from Medtronic pLc (Minneapolis, MN, USA) and Cook Medical LLC (Bloomington, IN, USA), plus a grant from Laboratoires Urgo S.A. (Chenôve, France). Andrew Bradbury had committee membership for the National Institute for Health Research Health Technology Assessment (HTA) Prioritisation Group and HTA Surgery Themed Call Board 2012–13, HTA Efficient Study Designs Board 2014–16, HTA Interventional Procedures Methods Group 2015–19 and HTA IP Panel 2015–19. In addition, Andrew Bradbury has received funding from STD Pharmaceutical Products Ltd (Hereford, UK) to travel to a foam sclerotherapy workshop in Tehran, Iran, in October 2016 and a grant to cover costs of undertaking a post-authorisation safety study in the UK and Europe. He also sat on the National Institute for Health and Care Excellence (NICE) committee for a clinical guideline (CG168) for the diagnosis and management of varicose veins. Nicky Cullum had committee membership on the HTA Commissioning Board from 2003 to 2008. David M Epstein has received grant funding from Vascular Insights LLC (Quincy, MA, USA) which was administered by the University of Granada. Alun H Davies has received grant funding from Medtronic, Vascular Insights, Laboratoires Urgo, Vascutek (Inchinnan, UK) and Actegy Health Ltd (Bracknell, UK), which are administered by Imperial College London. In addition, Alun H Davies has chaired the NICE clinical guideline (CG168) for the diagnosis and management of varicose veins.

Published May 2019 DOI: 10.3310/hta23240

Plain English summary

The EVRA RCT

Health Technology Assessment 2019; Vol. 23: No. 24 DOI: 10.3310/hta23240

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Venous leg ulcers are open wounds occurring on the legs of patients with venous disease. They are common, painful and distressing and reduce patient quality of life. Leg ulcers often result from valves in the leg veins not working properly. The valves normally force blood back up towards the heart; however, blood can flow backwards (reflux) when valves do not work properly, and this can cause swelling and ulceration. Compression therapy (wrapping bandages around the legs) has been shown to help ulcers heal, but it does not treat the underlying reflux problem with the veins. Newer, less invasive, techniques (known as endovenous ablation) have taken over from surgery to correct venous reflux and are more acceptable to patients as they can be performed quickly under local anaesthetic.

The aim of the trial was to find out if treating patients with leg ulcers by early endovenous ablation (within 2 weeks) and standard compression therapy can increase ulcer healing compared with standard compression therapy and delayed endovenous ablation once the ulcer has healed.

In total, 450 people agreed to take part in this study and were treated in 20 hospitals across England. Participants were randomly allocated to either early or delayed endovenous ablation and followed up for 12 months.

The trial found that treating the veins early resulted in quicker ulcer healing than delaying treatment until the ulcer had healed. The trial also showed that participants had more time without an ulcer if the treatment was performed early rather than after ulcer healing. No safety issues with early intervention were identified.

There is some evidence that quality of life was better in the early treatment group and that people in this group had less body pain. Treating ulcers early appears likely to be more cost-effective (i.e. a better use of NHS resources) than delayed treatment.

Future work will focus on collecting longer-term follow-up data to find out if early endovenous ablation also reduces the chances of the ulcer coming back.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Gohel *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton S016 7NS, UK.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.513

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 11/129/197. The contractual start date was in November 2016. The draft report began editorial review in May 2018 and was accepted for publication in September 2018. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Gohel *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Honorary Professor, University of Manchester, and Senior Clinical Researcher and Associate Professor, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk